Author's response to reviews

Title: Cost-effectiveness comparison between palpation- and ultrasound-guided thyroid fine-needle aspiration biopsies

Authors:

Ahmet Selçuk Can (selcukcan@endokrinoloji.com)

Version: 6 Date: 13 May 2009

Author's response to reviews:

The requested editorial change was made. Table 5 was changed from landscape format to portrait format. Few spelling errors were corrected. EndNote field codes were removed because there was a problem with extraction.